**Author’s response to reviews**

**Title:** Digital Media-based Health Intervention on the Promotion of Women's Physical Activity: A Quasi-Experimental Study

**Authors:**

Nooshin Peyman (peymann@mums.ac.ir)

Majid Rezai-Rad (mrezairad41@yahoo.com)

Hadi Tehrani (tehrani.hadi420@gmail.com)

Mahdi Gholian -Aval (gholianam@mums.ac.ir)

Mohammad Vahedian-Shahroodi (vahedianm@mums.ac.ir)

**Version:** 4  **Date:** 25 Dec 2017

**Author’s response to reviews:**

Editor Comments:

1- Please use initials to refer to each author's contribution in the 'Authors' Contributions' section, for example: "FC analyzed and interpreted the patient data regarding the hematological disease and the transplant. RH performed the histological examination of the kidney, and was a major contributor in writing the manuscript." Please also provide the statement 'All authors read and approved the final manuscript." at the end of this section.

R: Authors’ Contribution was revised

2- In line with ICMJE guidelines, BioMed Central requires registration of all clinical trials that are reported in manuscripts submitted to its journals. The ICMJE uses the World Health Organization (WHO) definition of a clinical trial, which is "any research study that prospectively assigns human participants or groups of humans to one or more health-related interventions to evaluate the effects on health outcomes.

If you have not already registered your study, we ask that you please do this now. The TRN, date of registration and the words 'retrospectively registered' should be included as the last line of the manuscript abstract.
R: We applied for registration this article at the Iranian Registry of Clinical Trials (IRCT). IRCT is a primary registry that participates in the WHO International Clinical Trials Registry Platform.

The TRN, date of registration and the words 'retrospectively registered added in the last line of the manuscript abstract.